Author's response to reviews

Title: Problems in early diagnosis of bladder cancer in spinal cord injury patients: Report of a case of simultaneous production of granulocyte colony stimulating factor and parathyroid hormone-related protein by squamous cell carcinoma of urinary bladder

Authors:

Subramanian Vaidyanathan (vaidyanathansiu@hotmail.com)
Paul Mansour (Paul.Mansour@mail.soh-tr.nwest.nhs.uk)
Munehisa Ueno (camtmune@saitama-med.ac.jp)
Kazuto Yamazaki (KazutoYamazaki@aol.com)
Meenu Wadhwa (mwadhwa@nibsc.ac.uk)
Bakul M Soni (Bakul.Soni@mail.soh-tr.nwest.nhs.uk)
Gurpreet Singh (Gurpreet.Singh@mail.soh-tr.nwest.nhs.uk)
Peter L Hughes (Peter.Hughes@mail.soh-tr.nwest.nhs.uk)
Ian D Watson (ian.watson@mail.soh-tr.nwest.nhs.uk)
Pradipkumar Sett (Pradipkumar.Sett@mail.soh-tr.nwest.nhs.uk)

Version: 2 Date: 25 Mar 2002

Dear Editor
I did not receive acknowledgement for this manuscript. Usually, BioMed Central is very prompt in acknowledging the receipt of submitted manuscripts. Perhaps, I made a mistake somewhere while submitting this manuscript in electronic format. Therefore, I am resubmitting the manuscript. I hope this manuscript reaches you safely now. I made alterations in the first two sentences of the abstract. I thank you and your editorial team for considering our submission for publication in your journal. I am grateful to you for valuable help.

With regards
Yours sincerely
Dr S Vaidyanathan